Author’s response to reviews

Title: National Health Insurance Accreditation pattern among private healthcare providers in Ghana

Authors:
Abena Agyeiwaa Lamptey (aaamoako79@yahoo.com)
Eric Nsiah-Boateng (nsiahboateng@yahoo.com)
Samuel Agyei Agyemang (sagyem85@gmail.com)
Moses Aikins (mksaikins@ug.edu.gh)

Version: 5 Date: 10 Jun 2017

Author’s response to reviews:

10 June 2017

Editorial Office
Archives of Public Health

Dear Editors:

Please find enclosed for your review, the revised manuscript entitled: “National health insurance accreditation pattern among private healthcare providers in Ghana”. The following are point-by-point responses to the reviewers/editorial comments:

1. The “Abstract” section is rewritten to reflect results of the confidence interval estimates of proportion of healthcare providers that passed the healthcare facility assessment; page 4, line 70-92.

2. The “Data analysis” subsection under the “Methods” section is updated to capture the confidence interval analysis for proportions of regions and proportions of healthcare providers that passed assessment of the accreditation process; page 9, line 200-203.

3. The “Healthcare provider performance score by region” subsection under the “Results” section is rewritten to capture the confidence interval estimates of the proportion of
4. The “Healthcare provider performance score by type of facility” subsection under the “Results” section is rewritten to capture the confidence interval estimates of the proportion of healthcare providers that passed the healthcare facility assessment; page 11, line 242-250.

5. The “Healthcare provider performance score by grade” subsection under the “Results” section is rewritten to capture the confidence interval estimates of the proportion of healthcare providers that passed the healthcare facility assessment; page 11-12, line 253-262.

6. The “Discussion section” is updated to capture results from the confidence interval estimates for the proportion of healthcare providers by region and facility type that passed the accreditation assessment; page 12-14, line 277-312.

7. The “Conclusions” section is updated to capture results of the confidence interval estimates; page 15, line 336-347.

8. Figure 2, 3 and 4 are updated to capture the confidence intervals estimates.

9. The “Reference” section is updated to reflect changes made in the Manuscript.

10. Since Table 1 describes the number of applications submitted for reapplication, confidence interval estimates were not determined for the results.

Thank you for considering this paper for publication. We look forward to your reply.

Sincerely,

----Signed...........

Moses Aikins, PhD

School of Public Health
College of Health Sciences
University of Ghana
Accra, Ghana